Author's response to reviews

Title: Partner randomized controlled trial: study protocol and coaching intervention

Authors:

Jane M Garbutt (jgarbutt@dom.wustl.edu)
Gabrielle Highstein (gabe.highstein@gmail.com)
Yan Yan (yany@wodosis.wustl.edu)
Robert C Strunk (strunk@kids.wustl.edu)

Version: 2 Date: 9 March 2012

Author's response to reviews: see over
March 9, 2012

Executive Editor  
*BMC Pediatrics*  
BioMed Central  
236 Gray's Inn Road  
London WC1X 8HB  
United Kingdom

Dear Ms. Neilan,

We are pleased to resubmit our manuscript “Partner randomized controlled trial: study protocol and coaching intervention.” We would like you to consider this original article for publication in *BMC Pediatrics* as a study protocol article. The article title has been changed for this submission.

All persons listed as authors are responsible for the reported research, have participated in the concept and design and drafting or revising of the manuscript, and have approved the manuscript s submitted. No manuscripts based on this protocol are under review by any other journal, nor have any been published elsewhere, and we confirm that the research was approved by the institutional review board at Washington University.

We are not aware of any conflicts of interest pertaining to the article and we certify that there are no affiliations with or involvement in any organization or entity with a direct financial interest in the subject matter discussed in the manuscript. Please address any further correspondence to:

J. Garbutt, MB, ChB,  
Department of Pediatrics  
Washington University School of Medicine, Box 8116  
660 South Euclid Avenue  
St. Louis, MO. 63110  
Phone: 314/454-8613  
Fax: 314/286/1149  
Email: jgarbutt@dom.wustl.edu

Sincerely,

Jane M. Garbutt, MB, ChB